Author's response to reviews

Title: Inflammatory fibroid polyp of the ileum presenting with small bowel obstruction in an adult patient: a case report

Authors:

Toygar Toydemir (toygartoydemir@hotmail.com)

Version: 3 Date: 30 April 2010

Author's response to reviews:

Dear editor, the revisions in the paper were marked with red.

For reviewer’s comment:
1- The type of resection was 'wedge resection', not a segmental resection. The submitted photograph is before resection so it is difficult to make a comment for the type of resection.
2- according to reviewer’s comment;'lack of any message for the improvement of clinical practice in the discussion', A paragraph for surgical management was added.
3- one more reference was added as ref. 9